Author's response to reviews

Title: Factors associated with adherence to Antiretroviral Therapy (ART) among adult people living with HIV and attending their clinical care, Eastern Ethiopia

Authors:

Shiferaw - Letta (samishife2006@yahoo.com)
Asrat - Demissie (asrat_dem@yahoo.com)
Lemessa - Oljira (Olemessa@yahoo.com)
Yadeta - Dessie (yad_de2005@yahoo.com)

Version: 8 Date: 19 October 2015

Author's response to reviews:

Dear Editor

Thank you for your comments once again and the following revision are made to our recent document.

1. Figure legend is placed next to the references
2. Tables are moved to the end of the manuscript (after the reference)
3. We have also made some editorials to the manuscript